Author's response to reviews

Title: Effect of medication review and cognitive behaviour treatment by community pharmacists of patients discharged from the hospital on drug related problems and compliance: design of a randomized controlled trial

Authors:

Abeer Ahmad (a.ahmad@vumc.nl)
Laura MC. Welschen (l.welschen@vumc.nl)
Jacqueline M. Dekker (jm.dekker@vumc.nl)
Giel Nijpels (g.nijpels@vumc.nl)
Jacqueline Hugtenburg (JG.Hugtenburg@vumc.nl)

Version: 2 Date: 5 March 2010

Author's response to reviews: see over
Cover letter

Changes made in version

- Trial registration number mentioned in abstract
- tables created with use of table tools
- Authors contributions further described
- reference 15 revised to a functioning website
- Competing interest changed, authors do not have competing interests

Minor revisions
1. title without capitalise
2. country added to each affiliations
3. abstract: section methods renamed
4. all text is presented in black
5. methods section renamed
6. the full citation details for reference 32
7. section literature renamed
8. tables and figures deleted from manuscript and uploaded as separate files.